Author's response to reviews

Title: The Dependency on Central Government Funding of Decentralised Health Systems: Experiences of the Challenges and Coping Strategies in the Kongwa District, Tanzania.

Authors:

GASTO FRUMENCE (frumencegasto@yahoo.co.uk)
TUMAINI NYAMHANGA (tumaininyamhanga@yahoo.com)
MUGHWIRA MWANGU (mmwangu@yahoo.com)
ANNA-KARIN HURTIG (anna-karin.hurtig@epiph.umu.se)

Version: 2 Date: 7 October 2013

Author's response to reviews:

Please note that this is a qualitative study, which has adhered to the RATS guidelines on qualitative research from the proposal development, data collection, analysis and writing of this manuscript.

We would like to inform you that our manuscript has now been edited by English language expert whose address is written below:

Proof-Reading-Service.com Ltd, Devonshire
Business Centre, Works Road, Letchworth Garden city, Hertfordshire, SG61GJ, United Kingdom: Office phone +44(0)2031500431,
Email: enquiries@proof-reading-service.com

Please be also notified that this manuscript has not been submitted anywhere, therefore, it is not under consideration for publication in any journal. The mentioning of journal of BMC Public Health in our previous covering letter was a human error and we have rectified it. We apologize for such inconvenience.